Author’s response to reviews

Title: Identification of differentially expressed lncRNAs and mRNAs in luminal-B breast cancer by RNA-sequencing

Authors:
Cheng-Liang Yuan (13568235628@163.com)
Xiang-Mei Jiang (412584237@qq.com)
Ying Yi (ying8975@163.com)
Jian-Fei E (1172200180@qq.com)
Nai-Dan Zhang (znd1984@126.com)
Xue Luo (54393816@qq.com)
Ning Zou (deylab@163.com)
Wei Wei (33911035@qq.com)
Ying-Ying Liu (dysrmyyliu@163.com)

Version: 4 Date: 12 Nov 2019

Author’s response to reviews:

Dear editors and reviewers,
Many thanks for your high efficiency of work and good suggestions. We have revised the manuscript according to the comments and the recommendations suggested by the editors and reviewers.

The responses to the editors’ comments are as follows:
1. We note that the current submission contains some textual overlap with other previously published works, particularly in the Background and Discussion sections. Please see the attached screenshots for details. While we understand that you may wish to express some of the same ideas contained in this publication, please be aware that we cannot condone the use of text from previously published work. If there is overlap in the Methods section, please ensure to summarize the methods and cite the source. Please re-phrase these sections to minimise overlap.
Response: Thanks for your comment. We have revised our manuscript to reduce the overlap.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant
tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.

Response: Thanks for your comment. A clean revised manuscript have been uploaded in the submission system.

Thank you again for your great help and attention. I am looking forward to hearing from you about the final decision.

Best regards and wishes!

Yours sincerely,

Ning Zou